\begin{table}
\caption{\normalsize{Multiwavelength properties of LINERs$^{\ast}$. } } \label{cha3:tab:results} % is used to refer this table in the text\\ % title of Table
\begin{tabular}{l l c c c c c c c c c c c c c}
\hline\hline % inserts double horizontal lines
Num& Name &X-ray &${\log(L_{\rm X})}$&{HST}& Broad &Radio &Stellar &UV &X-ray & UV \\
 & &Class. & (2--10~keV) & Class. &H${\alpha}$ &Comp. &Pop. (\%) &Var. & Var. & Comp. \\
(1)& (2) & (3) & (4) & (5) & (6) & (7) & (8) & (9) &(10) &(11)
\\ \hline
1 & NGC~0315\dotfill & AGN \dotfill& 41.768\dotfill&C\dotfill &Yes \dotfill &Yes J (1,3) \dotfill &3.8 \dotfill & \dotfill& \dotfill & \dotfill \\
2 & NGC~0410\dotfill &Non-AGN$^{\ast}$ \dotfill& 40.715\dotfill& \dotfill & \dotfill &No (1,5) \dotfill &2.7 \dotfill & \dotfill& \dotfill & \dotfill \\
3 & NGC~0474\dotfill &Non-AGN \dotfill&$<$38.478\dotfill& \dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
4 & IIIZW~035 \dotfill &Non-AGN \dotfill& 40.048\dotfill&C\dotfill & \dotfill &Yes F (7) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
5 & NGC~0524\dotfill &Non-AGN \dotfill& 38.586\dotfill&D\dotfill & \dotfill &No (8) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
6 & NGC~0833\dotfill & AGN \dotfill& 41.734\dotfill& \dotfill & \dotfill &No (8) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
7 & NGC~0835\dotfill & AGN \dotfill& 41.399\dotfill& \dotfill & \dotfill &Yes S (8) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
8 & NGC~1052\dotfill & AGN \dotfill& 41.240\dotfill&C\dotfill &Yes \dotfill &Yes (21) \dotfill &4.0 \dotfill & \dotfill& \dotfill &Yes (1)\dotfill \\
9 & NGC~2639\dotfill &Non-AGN$^{\ast}$ \dotfill&$<$38.277\dotfill&C\dotfill &Yes \dotfill &Yes F (9) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
10 & NGC~2655\dotfill & AGN$^{\ast}$\dotfill& 41.234\dotfill&C\dotfill & \dotfill &Yes S (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
11 & NGC~2681\dotfill & AGN \dotfill& 39.275\dotfill&C\dotfill &Yes \dotfill &No (1) \dotfill &4.2 \dotfill & \dotfill& \dotfill &Yes (1)\dotfill \\
12 & NGC~2685\dotfill & AGN$^{\ast}$\dotfill& 39.042\dotfill& \dotfill & \dotfill &No (1) \dotfill &1.4 \dotfill & \dotfill& \dotfill & \dotfill \\
13 & UGC~4881\dotfill &Non-AGN \dotfill&$<$38.369\dotfill&C\dotfill & \dotfill & \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
14 & 3C~218 \dotfill & AGN \dotfill& 42.081\dotfill& \dotfill & \dotfill &Yes J (10) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
15 & NGC~2787\dotfill & AGN \dotfill&$<$38.805\dotfill&C\dotfill &Yes \dotfill &Yes (1) \dotfill & \dotfill & \dotfill& \dotfill & No (1)\dotfill \\
16 & NGC~2841\dotfill & AGN \dotfill& 39.219\dotfill&C\dotfill & \dotfill & \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
17 & UGC~05101\dotfill & AGN \dotfill& 42.066\dotfill&C\dotfill & \dotfill &Yes S (7) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
18 & NGC~3185\dotfill &Non-AGN$^{\ast}$ \dotfill& 39.374\dotfill&C\dotfill & \dotfill &Yes marg.(9) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
19 & NGC~3226\dotfill & AGN \dotfill& 40.800\dotfill& \dotfill &Yes \dotfill &Yes F (5) \dotfill &2.2 \dotfill & \dotfill& \dotfill & \dotfill \\
20 & NGC~3245\dotfill & AGN \dotfill& 38.982\dotfill&C\dotfill & \dotfill &No (1) \dotfill &6.2 \dotfill & \dotfill& \dotfill & \dotfill \\
21 & NGC~3379\dotfill &Non-AGN \dotfill& 38.131\dotfill&C\dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
22 & NGC~3414\dotfill & AGN \dotfill& 39.863\dotfill& \dotfill & \dotfill &Yes F (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
23 & NGC~3507\dotfill &Non-AGN \dotfill&$<$37.204\dotfill&C\dotfill & \dotfill &No (1) \dotfill &32.6 \dotfill & \dotfill& \dotfill & \dotfill \\
24 & NGC~3607\dotfill &Non-AGN \dotfill& 38.765\dotfill&C\dotfill & \dotfill &Yes (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
25 & NGC~3608\dotfill &Non-AGN \dotfill& 38.205\dotfill&C\dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
26 & NGC~3623\dotfill &Non-AGN$^{\ast}$ \dotfill&$<$39.383\dotfill&C\dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
27 & NGC~3627\dotfill &Non-AGN$^{\ast}$ \dotfill& 39.415\dotfill&D\dotfill & \dotfill &Yes F (2,4) \dotfill &5.0(7.3)\dotfill & \dotfill& \dotfill & \dotfill \\
28 & NGC~3628\dotfill &Non-AGN \dotfill& 39.942\dotfill&U\dotfill & \dotfill &No Ext. (1,2) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
29 & NGC~3690B\dotfill & AGN \dotfill& 40.862\dotfill&C\dotfill & \dotfill &Yes S (11) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
30 & NGC~3898\dotfill &Non-AGN \dotfill&$<$38.776\dotfill&C\dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
31 & NGC~3945\dotfill & AGN \dotfill& 39.121\dotfill&C\dotfill & \dotfill &Yes (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
32 & NGC~3998\dotfill & AGN \dotfill& 41.319\dotfill&C\dotfill &Yes \dotfill &Yes (1) \dotfill &54.4 \dotfill &Yes \dotfill& \dotfill &Yes (2)\dotfill \\
33 & NGC~4036\dotfill & AGN \dotfill& 39.126\dotfill&C\dotfill &Yes \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
34 & NGC~4111\dotfill & AGN \dotfill&$<$40.364\dotfill&C?\dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill &No (1)\dotfill \\
35 & NGC~4125\dotfill & AGN \dotfill&$<$38.730\dotfill&C\dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
36 &IRAS~12112+0305\dotfill &Non-AGN$^{\ast}$ \dotfill& 41.226\dotfill& \dotfill & \dotfill &Yes F (7) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
37 & NGC~4261\dotfill & AGN \dotfill& 41.069\dotfill&U\dotfill & \dotfill &Yes J (1) \dotfill &2.2 \dotfill & \dotfill& \dotfill & \dotfill \\
38 & NGC~4278\dotfill & AGN \dotfill& 39.226\dotfill&C\dotfill &Yes \dotfill &Yes J (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
39 & NGC~4314\dotfill &Non-AGN \dotfill&$<$39.095\dotfill&C\dotfill & \dotfill &No (1) \dotfill &3.8 \dotfill & \dotfill& \dotfill &No (1)\dotfill \\
40 & NGC~4321\dotfill &Non-AGN \dotfill& 40.491\dotfill&C\dotfill & \dotfill &No Ext. S (5) \dotfill &24.7 \dotfill & \dotfill& \dotfill & \dotfill \\
41 & NGC~4374\dotfill & AGN \dotfill& 39.533\dotfill&C\dotfill & \dotfill &Yes J (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
42 & NGC~4410A\dotfill & AGN \dotfill& 41.176\dotfill&C\dotfill &Yes \dotfill &Yes (5) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
43 & NGC~4438\dotfill & AGN \dotfill&$<$39.047\dotfill&D\dotfill &Yes \dotfill &No (1) \dotfill &2.2 \dotfill & \dotfill& \dotfill &No (1)\dotfill \\
44 & NGC~4457\dotfill & AGN \dotfill& 38.814\dotfill& \dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
45 & NGC~4459\dotfill &Non-AGN \dotfill& 38.372\dotfill&C\dotfill & \dotfill &No (1) \dotfill &8.7 \dotfill & \dotfill& \dotfill & \dotfill \\
46 & NGC~4486\dotfill & AGN \dotfill& 40.823\dotfill&C\dotfill &Yes \dotfill &Yes (1) \dotfill & \dotfill &Yes \dotfill&Yes\dotfill &Yes (2)\dotfill \\
47 & NGC~4494\dotfill & AGN \dotfill& 38.777\dotfill&C\dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
48 & NGC~4552\dotfill & AGN \dotfill& 39.249\dotfill&C\dotfill & \dotfill &Yes (1) \dotfill & \dotfill &Yes \dotfill&Yes\dotfill &Yes (2)\dotfill \\
49 & NGC~4589\dotfill &Non-AGN \dotfill& 38.917\dotfill&C\dotfill & \dotfill &Yes (1,5) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
50 & NGC~4579\dotfill & AGN \dotfill& 41.175\dotfill&C\dotfill &Yes \dotfill &Yes (1) \dotfill & \dotfill &Yes \dotfill& \dotfill &Yes (2)\dotfill \\
51 & NGC~4596\dotfill &Non-AGN \dotfill& 38.467\dotfill&C\dotfill & \dotfill &No (1) \dotfill &4.1 \dotfill & \dotfill& \dotfill & \dotfill \\
52 & NGC~4594\dotfill & AGN \dotfill& 39.973\dotfill&C\dotfill & \dotfill &Yes (13) \dotfill & \dotfill &Yes \dotfill& \dotfill &Yes (2)\dotfill \\
53 & NGC~4636\dotfill &Non-AGN \dotfill&$<$39.026\dotfill&D\dotfill &Yes \dotfill &Yes (1) \dotfill & \dotfill & \dotfill& \dotfill & No (1)\dotfill \\
54 & NGC~4676A\dotfill &Non-AGN \dotfill& 39.855\dotfill&D\dotfill & \dotfill & \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
55 & NGC~4676B\dotfill & AGN \dotfill& 40.135\dotfill&C\dotfill & \dotfill & \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
56 & NGC~4698\dotfill & AGN \dotfill& 38.744\dotfill&C\dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
57 & NGC~4696\dotfill &Non-AGN \dotfill& 39.978\dotfill&C\dotfill & \dotfill &Yes F (14) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
58 & NGC~4736\dotfill & AGN \dotfill& 38.599\dotfill&C\dotfill & \dotfill &Yes (1) \dotfill &5.3 \dotfill &Yes \dotfill& \dotfill &Yes (1)\dotfill \\
59 & NGC~5005\dotfill & AGN$^{\ast}$\dotfill&$<$39.855\dotfill&C?\dotfill &Yes \dotfill &No (1) \dotfill &3.6 \dotfill & \dotfill&Yes\dotfill & \dotfill \\
60 & NGC~5055\dotfill & AGN \dotfill& 37.778\dotfill&C\dotfill & \dotfill &No (1) \dotfill &5.7(14.0)\dotfill & \dotfill& \dotfill &Yes (2)\dotfill \\
61 & MRK~266NE\dotfill & AGN \dotfill& 41.656\dotfill&C\dotfill & \dotfill & \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
62 & UGC~08696\dotfill & AGN \dotfill& 42.993\dotfill&C\dotfill & \dotfill &Yes F (7) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
63 & CGCG~162-010\dotfill &Non-AGN \dotfill& 41.426\dotfill&C\dotfill & \dotfill &Yes F (15) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
64 & NGC~5363\dotfill & AGN$^{\ast}$\dotfill& 39.782\dotfill& \dotfill & \dotfill &Yes J (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
65 & IC~4395\dotfill &Non-AGN$^{\ast}$ \dotfill&$<$40.797\dotfill& \dotfill & \dotfill & \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
66 &IRAS~14348$-$1447\dotfill &Non-AGN \dotfill& 41.737\dotfill&C\dotfill & \dotfill &Yes F (11) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
67 & NGC~5746\dotfill & AGN \dotfill& 40.223\dotfill&C?\dotfill & \dotfill &No (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
68 & NGC~5813\dotfill &Non-AGN \dotfill& 38.770\dotfill&C\dotfill & \dotfill &Yes (1) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
69 & NGC~5838\dotfill & AGN \dotfill& 39.206\dotfill&C\dotfill & \dotfill &Yes F (2) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
70 & NGC~5846\dotfill &Non-AGN \dotfill&$<$40.814\dotfill&C\dotfill & \dotfill &Yes J (1,2,4) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
71 & NGC~5866\dotfill &Non-AGN \dotfill& 38.296\dotfill& \dotfill & \dotfill &Yes F (1,2,4) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
72 & MRK~0848\dotfill &Non-AGN \dotfill& 41.145\dotfill&C\dotfill & \dotfill &Yes F (16) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
73 & NGC~6251\dotfill & AGN \dotfill& 41.583\dotfill&C\dotfill & \dotfill &Yes J (17,18) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
74 & NGC~6240\dotfill & AGN \dotfill& 42.408\dotfill&C\dotfill & \dotfill &Yes (19) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
75 &IRAS~17208$-$0014\dotfill & AGN \dotfill& 41.188\dotfill&C\dotfill & \dotfill &Yes F (7) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
76 & NGC~6482\dotfill &Non-AGN \dotfill& 39.335\dotfill& \dotfill & \dotfill &No (1) \dotfill &1.7\dotfill & \dotfill& \dotfill & \dotfill \\
77 & NGC~7130\dotfill & AGN \dotfill& 40.795\dotfill&C\dotfill & \dotfill &Yes S (8) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
78 & NGC~7285\dotfill & AGN$^{\ast}$\dotfill& 41.323\dotfill& \dotfill & \dotfill & \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
79 & NGC~7331\dotfill &Non-AGN \dotfill& 38.456\dotfill&C\dotfill & \dotfill &Yes F (3) \dotfill &1.9(2.1) \dotfill & \dotfill& \dotfill & \dotfill \\
80 & IC~1459\dotfill & AGN \dotfill& 40.511\dotfill&C\dotfill & \dotfill &Yes F (14) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
81 & NPM1G~$-$12.0625\dotfill &Non-AGN \dotfill&$<$41.463\dotfill&D\dotfill & \dotfill &Yes S (20) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
82 & NGC~7743\dotfill &Non-AGN$^{\ast}$\dotfill&$<$39.548\dotfill&C\dotfill & \dotfill &Yes F (9) \dotfill & \dotfill & \dotfill& \dotfill & \dotfill \\
\hline
\end{tabular}
\smallskip
{C-T}~=~Compton-Thick candidates. 1)
\citet{chiaberge_hubble_2005}; 2) \citet{maoz_murmur_2005}. Col.~7
Stellar populations younger than $10^{9}$~yr taken from
\citet{fernandes_stellar_2004} and
\citet{delgado_stellar_2004} (within parenthesis).\\
Col.~() Broad H${\alpha}$ reported by \citet{ho_search_1997}.\\
Col.~6 References: 1) \citet{nagar_radio_2005}; 2)
\citet{filho_nature_2000}; 3) \citet{filho_radio_2002}; 4)
\citet{filho_further_2004}; 5) \citet{filho_radio_2006}; 6)
\citet{anderson_size_2005}; 7) \citet{baan_radio_2006}; 8)
\citet{corbett_first_2002}; 9) \citet{ho_radio_2001}; 10)
\citet{taylor_vla_1990}; 11) \citet{condon_radio_1991}; 12)
\citet{smith_interstellar_2000}; 13)
\citet{bajaja_large-scale_1988}; 14)
\citet{slee_parsecscale_1994}; 15)
\citet{van_breugel_extended_1984}; 16)
\citet{clemens_modeling_2008}; 17) \citet{urry_unified_1995}; 18)
\citet{jones_high_1986}; 19) \citet{carral_15_1990}; 20)
\citet{sarazin_comparison_1995}; 21)
\citet{vermeulen_shroud_2003}.\\
Col.~13 Interacting types: 0~=~Isolate, 1~=~Merger, 1.5~=~Close
Interacting Pair, 2~=~Pair, 2.5~=~Wide Pair, 3~=~Triplet,
4~=~Compact Group, 4.5~=~1st group, 5~=~Group, 10~=~Cluster Center
and, 20~=~Cluster Member.
\end{table}